Author's response to reviews

Title: A compare between myocardial topical negative pressure levels of -25 mmHg and -50 mmHg in a porcine model

Authors:

Sandra Lindstedt (sandra.lindstedt@skane.se)
Per Paulsson (per.paulsson@skane.se)
Arash Mokhtari (arash.mokhtari@kane.se)
Bodil Gesslein (bodil.gesslein@med.lu.se)
Joanna Hlebowicz (joanna.hlebowicz@skane.se)
Malin Malmsjo (malin.malmsjo@med.lu.se)
Richard Ingemansson (richard.ingenansson@skane.se)

Version: 7 Date: 9 June 2008

Author's response to reviews:

We are delighted that you have accepted the article for publication.

The following formatting changes on your request have been done:

Title & Abstract are identical in the manuscript and in the submission system.

The authors’ qualifications have been removed from the manuscript and submissions system.

Affiliations have been written in full, in the following format: department, institute, city, and country.

Each email address has been written on a new line.

The following text has been removed;
'Article Type: Research Article'

Keywords
'Lindstedt et al.'

List of abbreviations is presented as a paragraph.

'Funding sources' has been moved to under the section heading 'Acknowledgements'.

Section headings have been changed as been requested.

Regards Sandra Lindstedt